Author's response to reviews

Title: Acute Low Back Pain is Marked by Variability: An Internet-Based Pilot Study

Authors:

Pradeep Suri (pradeep.suri@va.gov)
James Rainville (jrainvil@caregroup.harvard.edu)
Garrett M Fitzmaurice (fitzmaur@hsph.harvard.edu)
Jeffrey N Katz (jnkatz@partners.org)
Robert N Jamison (rjamison@partners.org)
Julia Martha (juli.martha@gmail.com)
Carol Hartigan (chartiga@caregroup.harvard.edu)
Janet Limke (jlimke@caregroup.harvard.edu)
Cristin Jouve (cjouve@caregroup.harvard.edu)
David J Hunter (david.hunter@sydney.edu.au)

Version: 4 Date: 3 October 2011

Author's response to reviews: see over
To: Editors, *BMC Musculoskeletal Disorders*

Ms. Ref. No.: 1840212398544790 (revision 3)
Title: Acute Low Back Pain is Marked by Variability: An Internet-Based Pilot Study

Dear Editors,

We have addressed all editorial production comments. We were uncertain how to resolve the matter of citation #6, because it has been formally accepted by the Clinical Journal of Pain and is now ‘In Press’. However, this status causes it to fall between the categories of ‘submitted’ and ‘publicly available’. We will defer to the opinion of the editorial production team.

Sincerely yours,

Pradeep Suri